ORIGINAL ARTICLE

EATING DISORDERS WILEY

Caregiver-reported household food insecurity and child-reported food insecurity in relation to eating disorder risk factors and symptoms among preadolescent children

Mikayla R. Barry PhD, MPH ^{1,2} 💿 📔 Kendrin R. Sonneville ScD, RD ¹ 💿	I
Andrea R. McGowan MPH ¹ Belinda L. Needham PhD, MA ² 💿	
Lindsay C. Kobayashi PhD, MSc ² 🔋 📔 Cindy W. Leung ScD, MPH ¹ 💿	

¹Department of Nutritional Sciences, University of Michigan School of Public Health, Ann Arbor, Michigan, USA ²Department of Epidemiology, University of Michigan School of Public Health, Ann Arbor, Michigan, USA

Correspondence

Mikayla R. Barry, University of Michigan School of Public Health, Ann Arbor, MI 48109, USA. Email: barrymik@umich.edu

Funding information

Eunice Kennedy Shriver National Institute of Child Health and Human Development, Grant/ Award Number: 4R00HD084758

Action Editor: Natasha L. Burke

Abstract

Objective: To examine the relation of caregiver-reported household food insecurity (FI) and child-reported FI with eating disorder (ED) risk factors and symptoms, including effect modification by gender, in preadolescent children.

Method: Data were from the Family Food Study, a cross-sectional study of households with incomes \leq 200% of the federal poverty line in southeastern Michigan. Children aged 8–10 years (n = 194) and their female primary caregivers reported separately on FI status. Children reported ED risk factors/symptoms via the 24-item Children's Eating Attitudes Test (ChEAT-24), with higher scores indicating more ED risk factors/symptoms. Linear mixed models were used to examine associations between FI measures with the ChEAT-24 total score, plus subscale scores for dieting, food preoccupation, weight preoccupation, vomiting, and social pressure to eat/gain weight. Models were adjusted for child age, child gender, caregiver race/ethnicity, caregiver education, and household income.

Results: Among all children, child-reported FI, but not caregiver-reported household FI, was associated with more ED risk factors/symptoms. Child-reported FI (vs. no FI) was associated with higher average ChEAT-24 total score ($\beta = 2.41$, 95% CI: 0.57, 4.25). Child-reported FI was also associated with more food preoccupation, more weight preoccupation, and more social pressure to eat. Caregiver-reported household FI was marginally associated with less dieting in girls, and child-reported FI was associated with more dieting in boys.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Discussion: Child-reported FI may be more salient than caregiver-reported household FI as a risk factor for ED-related outcomes in preadolescent children. Gender may modify the association between FI and dieting behavior.

Public Significance Statement: More child-reported food insecurity, but not parentreported household food insecurity, was associated with more eating disorder risk factors and symptoms among preadolescent boys and girls. These findings emphasize the need for future studies that investigate the role of food insecurity in the development of eating disorders, especially studies that measure child-reported experience of food insecurity.

KEYWORDS

children, feeding and eating disorders, food insecurity, gender identity

1 | INTRODUCTION

Preadolescence (i.e., the developmental stage immediately preceding adolescence, from approximately 8-11 years of age) may be a critical period for the prevention of eating disorders (EDs). While the median age of onset for full-syndrome EDs is in mid-to-late adolescence (Hanson et al., 2007), proximal risk factors (e.g., body weight and shape concerns, dieting, and so forth) and subthreshold behavioral symptoms (e.g., fasting, purging, binge eating, and so forth) frequently emerge in preadolescence (Bernier et al., 2010; Combs et al., 2013; Swanson et al., 2014). Preadolescent children with ED risk factors/symptoms are more likely to develop full-syndrome EDs compared to preadolescents without ED risk factors/symptoms (McClelland et al., 2020; Swanson et al., 2014). Subthreshold ED symptoms are also associated with increased risk of several health and developmental concerns later in life, such as more psychological distress, worse self-rated health, more binge drinking and drug use, and metabolic syndrome, to name a few (Kärkkäinen et al., 2018; Swanson et al., 2014; Tanofsky-Kraff et al., 2012). A better understanding of the etiology of ED risk factors/ symptoms in preadolescence is urgently needed to inform prevention efforts that can be implemented before prolonged disordered eating and/or ED onset occur. However, little is known about how environmental factors, such as the home food environment, are related to ED risk factors/symptoms in preadolescent children.

Exposure to food insecurity is one factor that might increase the risk for ED-related outcomes among preadolescents. Food insecurity (FI) is defined as a lack of sufficient financial resources to obtain enough quality food (Anderson, 1990). In 2020, approximately one in six U.S. children lived in food insecure households (Coleman-Jensen et al., 2021). There are multiple mechanisms through which FI might increase the risk for ED-related outcomes in children. First, worry about FI might directly increase food preoccupation. Indeed, an association between FI (vs. no FI) and more food preoccupation has been found in some studies of adults (Barry et al., 2021; Poll et al., 2020). In addition, FI might also increase ED risk indirectly through several possible pathways: FI (vs. no FI) in preadolescence has been associated with elevated depressive symptoms (Thomas et al., 2019) and greater

adiposity in children (Burke et al., 2016; Jansen et al., 2017; Lee et al., 2018). Experiences of household FI are also associated with some parental attitudes and behaviors, like more concern about child weight and more use of restrictive and pressured feeding practices (Bauer et al., 2015; Conlon et al., 2015). In turn, the aforementioned child and parental characteristics have been associated with greater likelihood of various ED risk factors/symptoms in children/adoles-cents, including dieting/food restriction, concern about body weight/ body dissatisfaction, use of extreme weight control behaviors (which can include self-induced vomiting), and higher scores on scales of overall disordered eating (Jendrzyca & Warschburger, 2016; Larsen et al., 2015; Loth et al., 2014; Stice et al., 2011).

Preadolescents could be especially sensitive to the negative consequences of FI. While in the midst of a critical period for the development of eating habits and body image, preadolescents are also old enough to be cognizant of FI in the home (Fram et al., 2011; Leung et al., 2020). However, few studies have investigated this association in samples that include preadolescents: Masler et al. (2021) found that adult-reported household FI (vs. no FI) was associated with more weight loss attempts and unhealthy weight control behaviors in a sample of children aged 8–15 years. Another study by Altman et al. (2019) found that children in grades 4–8 (approximate ages 9–14) with self-reported FI were more likely to experience body dissatisfaction compared to those with no FI. While both of these studies included preadolescent children, the wide age ranges did not allow for specific examination of this key developmental window.

In addition, previous research suggests that while child and caregiver reports of FI are correlated, FI reports may disagree for over one-third of child-caregiver pairs (Bernard et al., 2018; Carlos Chavez et al., 2017). Although adult-reported household FI is the most commonly used method of assessing FI in the literature, child self-report more accurately reflects children's perceptions and experiences of FI (Fram et al., 2013). Understanding whether caregiver and child report of FI differ in their relationship to ED risk factors/symptoms is important for informing future research and FI assessment methods for both FI and ED prevention/intervention efforts. However, to the best of our knowledge,

EATING DISORDERS-WILEY 1333

previous studies on the association between FI and ED risk factors/ symptoms have not included measures of both caregiver- and childreported FI.

Finally, women have a higher lifetime prevalence of EDs compared to men (Hudson et al., 2007); there is also evidence that more FI is related to higher weight among girls but not boys (Burke et al., 2016; Jansen et al., 2017), and one study found that while more FI is associated with more ED risk factors/symptoms in both adult males and females, the association is stronger for males (Barry et al., 2021). Understanding whether gender modifies the associations between FI and ED-related outcomes in preadolescents may inform how potential FI and ED prevention/intervention programs could be targeted to the subgroups of children who would benefit the most. Yet, to the best of our knowledge, only one study has examined gender differences in the association between FI and ED-related outcomes in preadolescents. This previous study found that the association between FI and body dissatisfaction did not vary by sex (Altman et al., 2019).

Therefore, the goal of the present study is to assess the association of caregiver-reported household FI and child-reported FI with ED risk factors/symptoms, including the potential modifying role of gender, among preadolescent children.

2 | METHODS

2.1 | Participants

This study uses cross-sectional data from baseline assessments in the Family Food Study (FFS) that were conducted from September 2018 to December 2019. The FFS is a cohort of children aged 8-10 years at baseline, and their female primary caregivers. Families from southeastern Michigan with household incomes ≤200% of the federal poverty line whose primary spoken language was English were recruited to participate via convenience sampling through the University of Michigan Health Research website, social media outreach, and flyer posting in neighborhood community centers. More than one child per household could participate when multiple children were age-eligible. Data collection occurred in participants' homes, where children and their female primary caregivers completed various assessments, including anthropometric measures, cognitive tasks, and in-depth survey questionnaires. The study focused on female caregivers because they generally report greater awareness of the household food environment and FI than male caregivers (Flagg et al., 2014; Nagao-Sato et al., 2021). All measures used in this study were collected through computer-based surveys that were self-administered separately by children and caregivers. Children were encouraged to ask trained staff members for assistance as needed, which included clarifying the meaning of questions and providing reading assistance for younger children. Caregivers provided written informed consent and children verbally assented to participate in the study. Female primary caregivers were compensated \$30 (U.S. dollars), and children received small toy prizes. This study was approved by the University of Michigan Medical School Institutional Review Board.

2.2 | Measures

2.2.1 | Food insecurity

Household FI was reported by female primary caregivers using the 18-item United States Department of Agriculture (USDA) Household Food Security Survey Module (Economic Research Service, 2012), which is known to have excellent validity and reliability (Margues et al., 2015). This survey asked questions about the household's ability to afford food over the previous 12 months and how the ability to afford food affected the quality or quantity of food available to children and adults in the household. Responses were scored on a scale of 0-18, where higher scores equate to more FI. Scores were grouped into two categories: no household FI (score 0-2) and household FI (score \geq 3) (ERS, 2012). Household FI indicates that the food quality and/or quantity in the household is impaired due to lack of money to buy food (ERS, 2012). In our sample, internal consistency measured by Chronbach's α was .91 for the USDA Household Food Security Survey Module, which is consistent with previous studies (Margues et al., 2015).

Children reported their food security status via the five-item Child Food Security Assessment (CFSA), which is validated for use in children as young as 6 years old (Fram et al., 2013). This survey asked about the child's experiences related to FI, such as worry about getting enough food to eat or experiencing hunger because there is not enough food to eat. Children reported how frequently they had these experiences by choosing "never," "sometimes," or "a lot." Responses were scored on a scale of 0–10, where higher scores equate to more FI. Consistent with a previous study, scores were also grouped into two categories: no child-reported FI (score 0–1) and child-reported FI (score \geq 2) (Landry et al., 2019b). Chronbach's α for the CFSA was .73 in our sample.

2.2.2 | ED risk factors/symptoms

ED risk factors/symptoms were self-reported by children using the Children's Eating Attitudes Test (ChEAT) (Maloney et al., 1988). This questionnaire is designed to capture ED risk factors/symptoms among children 8-13 years old (Maloney et al., 1988). Children reported how frequently they experienced different risk factors/symptoms using six response options that each equated to a numeric score ranging from 0 to 3: "never" (score of 0), "rarely" (0), "sometimes" (0), "often" (1), "very often" (2), or "always" (3). Response scores were summed to produce a total ChEAT score for each child. The reliability and validity of the ChEAT have been shown in various studies (Lommi et al., 2020; Maloney et al., 1988; Rojo-Moreno et al., 2011; Smolak & Levine, 1994). Consistent with previous research, responses for two of the original 26 ChEAT items were dropped from this study: #19 ("I can show self-control around food") and #25 ("I enjoy trying new rich foods") (Lommi et al., 2020; Maloney et al., 1988; Murphy et al., 2019; Smolak & Levine, 1994). The resulting ChEAT-24 total score had a possible range of 0-72, where higher scores indicate more

TABLE 1 Questionnaire items from the original 26-item Children's Eating Attitudes Test (ChEAT-26)^a that comprise the ChEAT-24 and each of the ChEAT subscale scores^b

		Subscale (possible score range)					
ltem ^c		Dieting (0-9)	Food preoccupation (0–9)	Weight preoccupation (0–9)	Vomiting (0–6)	Social pressure (0–9)	
1	I am scared about being overweight			Х			
2	I stay away from eating when I am hungry						
3	I think about food a lot of the time		Х				
4	I have gone on eating binges where I feel that I might not be able to stop		Х				
5	I cut my food into small pieces						
6	I am aware of the energy (calorie) content in foods that I eat	х					
7	I try to stay away from foods such as breads, potatoes, and rice						
8	I feel that others would like me to eat more					х	
9	I vomit after I have eaten				Х		
10	I feel very guilty after eating						
11	I think a lot about wanting to be thinner			Х			
12	I think about burning up energy (calories) when I exercise						
13	Other people think I am too thin					х	
14	I think a lot about having fat on my body			Х			
15	I take longer than others to eat my meals						
16	I stay away from foods with sugar in them	Х					
17	I eat diet foods	Х					
18	I think that food controls my life		Х				
20	I feel that others pressure me to eat					Х	
21	I give too much time and thought to food						
22	I feel uncomfortable after eating sweets						
23	I have been dieting						
24	I like my stomach to be empty						
26	I have the urge to vomit after eating				х		

^aThe ChEAT-26 was designed by Maloney et al. (1988).

^bChEAT subscales recommended by Murphy et al. (2019).

^cltem numbers based on order numbers from the original ChEAT-26. ChEAT-26 items 19 and 25 are excluded from the ChEAT-24.

ED risk factors/symptoms. In the present sample, Chronbach's α was .74 for the ChEAT-24.

To examine specific ED risk factors/symptoms, we also used a set of continuous ChEAT subscale scores recommended by Murphy et al. (2019): dieting, food preoccupation, weight preoccupation, vomiting, and social pressure to eat or gain weight. The questionnaire items that comprise the ChEAT-24 and its subscales, as well as possible ranges for the subscale scores, are shown in Table 1.

2.2.3 | Covariates

Children self-reported their age and gender. Female primary caregivers reported their own race/ethnicity, highest level of education, annual household income, and the number of people in their household. Household income and household size were used to calculate the percentage of the poverty line based on the United States Department of Health and Human Services (HHS) Federal Poverty Guidelines for 2019 (Department of Health and Human Services, 2019).

2.3 | Statistical analysis

For survey measures with multiple items, participants who did not complete all items were considered missing for that measure. Children with non-missing data for at least one FI exposure and at least one ChEAT-24 measure (total score or a subscale score) were included in the analysis. Only one child was excluded due to missing data for all

EATING DISORDERS-WILEY 1335

ChEAT-24 measures. Among the analytic sample of 194 children, there were no children with missing data for caregiver-reported household FI, child-reported FI, or covariates. The number of children with non-missing data for each outcome were as follows: ChEAT-24 total score, 186; dieting subscale, 192; food preoccupation subscale, 193; weight preoccupation subscale, 194; vomiting subscale, 192; and social pressure subscale, 193. The analytic sample came from 166 households: 139 households with one participating child,

TABLE 2 Sociodemographic characteristics by 24-item Children's Eating Attitudes Test (ChEAT-24) total score and ChEAT subscale scores among preadolescent children aged 8–10 years old from low-income households in southeastern Michigan

			ChEAT-24 Total score	Diet subs	ing scale	Food preoccu subscale	pation	Weight preoccu subscale	oation	Vom subs	iting cale	Soci subs	al pressu scale	ıre
	N ^a	Percentage	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
Overall	194	100	5.2	6.2	0.8	1.3	0.8	1.5	1.1	2.1	0.2	0.7	0.5	1.6
Child age (years)														
8 ^b	66	34.0	4.6	6.5	0.5	1.0	1.1	1.8	0.6	1.4	0.2	0.8	0.4	1.4
9	72	37.1	4.8	4.8	0.8	1.3	0.5	0.9	1.4	2.2	0.2	0.8	0.4	1.1
10 ^c	56	28.9	6.4	7.4	1.0	1.5	0.7	1.5	1.5	2.6	0.1	0.5	0.8	2.2
p ^d			.24		.11		.08		.02		.71		.31	
p, trend ^e			.11		.04		.11		.01		.42		.19	
Child gender														
Воу	90	46.4	5.0	6.5	0.7	1.3	0.7	1.2	1.1	2.1	0.1	0.5	0.7	1.9
Girl	104	53.6	5.3	6.0	0.8	1.3	0.8	1.7	1.1	2.2	0.2	0.8	0.4	1.3
p			.72		.63		.61		.97		.39		.28	
Caregiver race/ethnicity	,													
White	115	59.3	4.4	5.5	0.6	1.1	0.7	1.5	1.0	2.1	0.1	0.7	0.4	1.3
Caregiver of color ^f	79	40.7	6.4	7.1	1.0	1.5	0.8	1.5	1.3	2.2	0.2	0.7	0.7	2.0
р			.04		.03		.75		.34		.58		.15	
Caregiver education														
≤High school diploma	31	16.0	4.3	5.5	0.8	1.2	0.8	1.6	1.1	2.1	0.3	0.9	0.4	1.1
Some college	100	51.6	5.9	7.3	0.7	1.3	0.7	1.6	1.3	2.2	0.1	0.6	0.6	1.8
≥Bachelor's degree	63	32.5	4.4	4.4	0.8	1.4	0.8	1.3	0.9	1.9	0.2	0.8	0.5	1.4
р			.23		.91		.93		.38		.54		.69	
Household income (% o	f pove	rty line) ^g												
<50	43	22.1	6.5	6.4	1.3	1.6	1.0	2.0	1.1	2.0	0.3	0.9	0.7	1.5
50 to <100	48	24.7	6.0	7.5	0.8	1.4	0.6	1.1	1.4	2.1	0.1	0.8	0.8	1.9
100 to <150	53	27.3	4.9	6.4	0.6	1.1	0.7	1.6	1.2	2.3	0.2	0.7	0.5	1.8
≥150	50	25.8	3.7	4.2	0.5	1.0	0.7	1.2	0.9	2.0	0.0	0.3	0.2	0.9
р			.16		.03		.75		.67		.22		.34	
p, trend			.03		.004		.61		.51		.07		.11	
Caregiver-reported hour	sehold	food insecu	rity											
No food insecurity	78	40.2	5.2	5.5	0.8	1.2	0.7	1.2	1.0	2.0	0.2	0.8	0.4	1.2
Food insecurity	116	59.8	5.2	6.7	0.7	1.3	0.8	1.6	1.2	2.2	0.2	0.6	0.6	1.8
р			.98		.55		.69		.46		.74		.30	
Child-reported food inse	ecurity	score												
0	39	20.1	4.0	5.7	0.8	1.3	0.2	0.6	0.9	2.0	0.2	0.9	0.4	1.4
1	45	23.2	3.9	4.1	0.8	1.2	0.6	1.2	0.7	1.7	0.1	0.5	0.3	1.4
2-3	51	26.3	4.9	6.2	0.6	1.1	0.5	1.0	1.3	2.2	0.2	0.6	0.5	1.4
4-5	29	15.0	7.4	5.8	0.5	1.1	1.9	2.0	1.9	2.8	0.2	0.7	0.4	1.1
6-10	30	15.5	7.0	8.8	1.2	1.8	1.0	2.1	1.1	1.9	0.2	0.8	1.3	2.4

TABLE 2 (Continued)

			ChEAT-24 Total score	Dieting 1 subscale		Food Dieting preoccupation subscale ubscale		Weight preoccupation subscale		Vomiting subscale		Social pressure subscale		ıre
	N ^a	Percentage	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
p			.04		.24		<.001		.17		.95		.09	
p, trend			.005		.63		<.001		.12		.73		.03	

Abbreviation: SD, standard deviation.

^aDue to missing values, *n* varied for each outcome. Among the total analytic sample of 194 children, the specific *n* values for each outcome were as follows: ChEAT-24 total score, 186; dieting subscale, 192; food preoccupation subscale, 193; weight preoccupation subscale, 194; vomiting subscale, 192; social pressure subscale, 193.

^bIncludes two children who were 7 years old.

^cIncludes two children who were 11 years old.

^dBased on *F* test from unadjusted linear mixed models where the sociodemographic characteristic was modeled categorically and ChEAT total and subscale scores were modeled as continuous outcomes. Models accounted for clustering of children within the same households.

^eBased on Wald test from unadjusted linear mixed models where the sociodemographic characteristic was modeled ordinally and ChEAT total and subscale scores were modeled as continuous outcomes. Every model included a random intercept for household identifier to account for clustering of children within the same households.

^fAmong the 79 children with a caregiver of color, 53 (67.1%) had a Black/African American caregiver, three (3.8%) had a Hispanic caregiver, one (1.3%) had a Middle Eastern/North African caregiver, two (2.5%) had an Asian caregiver, 11 (13.9%) had a multi-racial/multi-ethnic caregiver, and nine (11.4%) had a caregiver of other race/ethnicity.

^gBased on the U.S. Department of Health and Human Services (DHHS) Federal Poverty Guidelines for 2019 (Department of Health and Human Services, 2019).

26 households with two participating children, and one household with three participating children.

We used linear mixed regression models with children (Level 1) clustered within households (Level 2) to estimate the mean difference in ChEAT-24 total scores and subscale scores by caregiverreported household FI and, in separate models, by child-reported FI. Every model included a random intercept for household identifier to account for clustering of children within the same households. Caregiver-reported household FI was modeled categorically (household FI vs. no household FI) according to standard practice (ERS, 2012). Child-reported FI was modeled both categorically and continuously, consistent with previous studies (Fram et al., 2015; Landry et al., 2019b). All models were adjusted for the following: child age (in whole years, continuous), child gender (boy vs. girl), caregiver race/ethnicity (white vs. caregiver of color), caregiver highest level of education (three categories: high school diploma or less, some college without a bachelor's degree, bachelor's degree or more), and household income as percentage of the poverty line (continuous). Caregiver race/ethnicity and education were collapsed into two and three categories, respectively, to preserve statistical power. We adjusted for child age, caregiver race/ethnicity, caregiver education, and household income because each of these variables may predict both FI and ED risk, thus confounding the associations of interest. Caregiver race/ethnicity, rather than child race/ethnicity, was used because we hypothesize that caregiver race/ethnicity may be more strongly associated with experiences of structural racism (e.g., housing and employment discrimination, incarceration, and so forth), which may lead to FI (Odoms-Young & Bruce, 2018). We additionally adjusted for child gender as an independent predictor of ED risk, to optimize statistical precision.

Interaction terms between each FI variable and child gender were used to test for effect modification by gender in each of the adjusted models. When the interaction term was statistically significant, results were additionally stratified by gender. All statistical analyses were conducted using SAS version 9.4. Statistical significance was determined using an alpha level of .05.

3 | RESULTS

Sociodemographic characteristics and their relation to the ChEAT-24 total score and subscale scores can be seen in Table 2. Children had a mean age of 8.9 (SD = 0.8) years, and 46.4% were boys. A large proportion (40.7%) of the sample had a female primary caregiver of color. Specifically, for the 79 children with a caregiver of color, caregivers were Black/African American (67.1% of children), Hispanic (3.8%), Middle Eastern/North African (1.3%), Asian (2.5%), multi-racial/multiethnic (13.9%), and other race/ethnicity (11.4%). Due to the recruitment criteria, 46.9% of all children lived in a household with a total income below the poverty line. The majority (59.8%) of children lived in food-insecure households based on caregiver report. Similarly, more than half of children (56.7%) perceived themselves as food insecure, with a mean child-reported FI score of 2.6 (SD = 2.3) and a range of 0-10. Among all children, 122 (62.9%) reported the same food security status as their caregiver, 33 (17.0%) reported FI when their caregiver reported no FI in the home, and 39 (20.1%) reported no FI when their caregiver reported FI in the home. Children had a mean ChEAT-24 total score of 5.2 (SD = 6.2), with total ChEAT-24 scores ranging from 0 to 36. In bivariate analyses, older child age, having a caregiver of color (vs. a white caregiver), lower household income, and higher

TABLE 3Results from linear mixedmodels examining the association ofcaregiver-reported household foodinsecurity (vs. no household foodinsecurity) with 24-item Children's EatingAttitudes Test (ChEAT-24) total andsubscale scores among preadolescentchildren aged 8–10 years old from low-income households in southeasternMichigan

	Unadjuste	ed estimates ^a	Adjusted	estimates ^b
	β	95% CI	β	95% CI
ChEAT-24 total score	-0.02	(-1.91, 1.87)	-0.06	(-2.01, 1.90)
Dieting subscale	-0.11	(-0.49, 0.26)	-0.11	(-0.50, 0.27)
Food preoccupation subscale	0.09	(-0.35, 0.52)	0.09	(-0.36, 0.54)
Weight preoccupation subscale	0.23	(-0.38, 0.85)	0.19	(-0.45, 0.83)
Vomiting subscale	-0.03	(-0.24, 0.17)	-0.05	(-0.26, 0.16)
Social pressure subscale	0.24	(-0.22, 0.70)	0.19	(-0.29, 0.67)

^aEstimates from separate linear mixed models where the ChEAT-24 total and subscale scores were modeled as continuous outcomes. Every model included a random intercept for household identifier to account for clustering of children within the same households. Each β value represents the mean difference in the ChEAT-24 total score or subscale score for children with caregiver-reported household food insecurity versus children with no household food insecurity.

^bAdjusted for child age, child gender, caregiver race/ethnicity, caregiver education, and household income as percentage of the poverty line.

TABLE 4Results from linear mixed models examining the association of child-reported food insecurity (modeled both categorically and
continuously) with the 24-item Children's Eating Attitudes Test (ChEAT-24) total and subscale scores among preadolescent children aged 8–
10 years old from low-income households in southeastern Michigan

	Unadjusted esti	mates ^a	Adjusted estima	ites ^b
	β	95% CI	β	95% CI
Child-reported food insecurity (vs. child rep	port of no food insecurity	c		
ChEAT-24 total score	2.26	(0.47, 4.05)	2.41	(0.57, 4.25)
Dieting subscale	-0.08	(-0.45, 0.30)	-0.07	(-0.45, 0.32)
Food preoccupation subscale	0.60	(0.19, 1.02)	0.55	(0.10, 0.99)
Weight preoccupation subscale	0.62	(0.02, 1.23)	0.71	(0.08, 1.34)
Vomiting subscale	0.03	(-0.17, 0.23)	-0.01	(-0.22, 0.20)
Social pressure subscale	0.34	(-0.12, 0.79)	0.35	(-0.13, 0.83)
Child-reported food insecurity score (per 1	-unit higher score) ^d			
ChEAT-24 total score	0.54	(0.16, 0.92)	0.61	(0.21, 1.01)
Dieting subscale	0.02	(-0.06, 0.10)	0.03	(-0.05, 0.11)
Food preoccupation subscale	0.15	(0.07, 0.24)	0.14	(0.05, 0.24)
Weight preoccupation subscale	0.08	(-0.04, 0.21)	0.11	(-0.03, 0.24)
Vomiting subscale	0.01	(-0.03, 0.05)	0.00	(-0.05, 0.04)
Social pressure subscale	0.12	(0.03, 0.22)	0.15	(0.04, 0.25)

^aEstimates from separate linear mixed models where the ChEAT-24 total and subscale scores were modeled as continuous outcomes. Every model included a random intercept for household identifier to account for clustering of children within the same households. When child-reported food insecurity is modeled categorically, each β value represents the mean difference in the ChEAT-24 total score or subscale score for children with self-reported food insecurity versus children who self-reported no food insecurity.

^bAdjusted for child age, child gender, caregiver race/ethnicity, caregiver education, and household income as percentage of the poverty line.

^cWhen child-reported food insecurity is modeled categorically, each β value represents the mean difference in the ChEAT-24 total score or subscale score for children with self-reported food insecurity vs. children who self-reported no food insecurity.

^dWhen child-reported food insecurity is modeled continuously, each β value represents the mean difference in the ChEAT-24 total score or subscale score per 1-unit increase in child-reported food insecurity score.

child-reported FI score were each associated with higher scores for one or more ChEAT-24 measures. Child gender, caregiver education, and caregiver-reported household FI were not associated with any ChEAT-24 measures in bivariate analyses. In adjusted analyses, caregiver-reported household FI (vs. no household FI) was not associated with the ChEAT-24 total score or any subscale scores (Table 3). For the association between caregiverreported household FI and dieting, there was evidence for effect 1338 WILEY-EATING DISORDERS

modification by gender (p-interaction = .04). Gender-stratified analyses identified a null association for boys ($\beta = 0.26$, 95% CI: -0.40, 0.92). For girls, there was an inverse association, where caregiverreported household FI (vs. no household FI) was associated with less dieting, which did not reach statistical significance ($\beta = -0.45$, 95%) CI: -0.93, 0.02). None of the other examined associations involving caregiver-reported household FI were modified by gender (effect modification analyses not shown in tables).

In contrast, children who self-reported FI had a mean ChEAT-24 total score that was 2.41 units higher (95% CI: 0.57, 4.25) compared to children who reported no FI. Child-reported FI (vs. child report of no FI) was also associated with more food preoccupation ($\beta = 0.55$, 95% CI: 0.10, 0.99) and more weight preoccupation (β = 0.71, 95% CI: 0.08, 1.34). However, child-reported FI (vs. child report of no FI) was not associated with dieting, vomiting, or social pressure to eat/gain weight (Table 4). When modeled categorically, none of the examined associations involving child-reported FI were modified by gender (data not shown).

When child-reported FI score was modeled continuously, each 1-unit increase in child-reported FI score was associated with a 0.61-unit increase (95% CI: 0.21, 1.01) in mean ChEAT-24 total score. Higher child-reported FI score was also associated with more food preoccupation ($\beta = 0.14, 95\%$ CI: 0.05, 0.24) and more social pressure to eat/gain weight ($\beta = 0.15, 95\%$ CI: 0.04, 0.25). Continuous child-reported FI score was not associated with dieting, weight preoccupation, or vomiting in the overall sample (Table 4). However, the association between child-reported FI score and dieting was modified by gender (p-interaction = .002). In gender-stratified analyses, higher child-reported FI score was associated with more dieting among boys $(\beta = 0.16, 95\%$ CI: 0.04, 0.29) but was not associated with dieting among girls ($\beta = -0.06$, 95% CI:-0.16, 0.04). None of the other examined associations involving the continuous measure of child-reported FI were modified by gender (data not shown).

DISCUSSION 4

To the best of our knowledge, this is the first study to examine how both caregiver- and child-reported FI are associated with ED-related outcomes among low-income preadolescent children. We found that child-reported FI, but not caregiver-reported household FI, was associated with more ED risk factors/symptoms. Specifically, more childreported FI was associated with higher ChEAT-24 total score, more food preoccupation, more weight preoccupation, and more social pressure to eat or gain weight.

The associations of child-reported FI with a higher mean ChEAT-24 total score and more food preoccupation were robust regardless of whether child-reported FI was modeled continuously or categorically. These findings are consistent with previous literature in adults suggesting an association of self-reported FI with higher composite measures of disordered eating (Barry et al., 2021; Becker et al., 2019; El Zein et al., 2019; Laraia et al., 2015) and with more food preoccupation (Barry et al., 2021; Poll et al., 2020). The association of child-

reported FI (vs. child report of no FI) with more weight preoccupation is consistent with the study by Altman et al. (2019) who found an association between child-reported FI (vs. no FI) and more body dissatisfaction among children in grades 4-8. In addition, we found that when modeled continuously, child-reported FI was associated with more social pressure to eat or gain weight. This finding corroborates prior studies showing that parents in food insecure households are more likely to be concerned with child weight (Bauer et al., 2015) and to use pressured feeding practices (Conlon et al., 2015). Our finding suggests that children who report FI may sense this parental pressure. There was no association between child-reported FI and vomiting in this study, which is consistent with a previous study in adolescents (Hooper et al., 2020) but in contrast to findings in adults (Barry et al., 2021; Becker et al., 2019). It may be that self-induced vomiting is too rare to study in non-adult samples.

Overall, our results reflect small but clinically meaningful differences in ED risk factors/symptoms according to child-reported FI. For example. Lommi et al. (2020) found that among Finnish children (mean age of 11.6 years), those who screened positive for disordered eating symptoms based on a validated guestionnaire had a mean ChEAT-24 total score that was 4.76 units higher compared to children who screened negative for disordered eating symptoms. These results suggest that a 2.41-unit increase in ChEAT-24 total score-the average difference between children with and without self-reported FI in our study-could explain more than half of the difference between screening positive versus screening negative for disordered eating symptoms.

Our results suggest that child-reported FI may be a more salient risk factor for ED-related outcomes than caregiver-reported household FI. Similar to Bernard et al. (2018), we found that 17.0% of children reported FI when their caregiver reported no household FI, and 20.1% reported no FI when their caregiver reported household FI. There are many reasons why a child's self-report of FI may differ from their caregiver's report. First, caregivers may be unaware of the full extent to which children perceive and are affected by FI (Bernard et al., 2018; Fram et al., 2011), or caregivers might be embarrassed to report the full extent of FI in their household (Middleton et al., 2018), while their children might be more open about FI. On the other hand, some caregivers strive to shield their children from full awareness of FI (Fitchen, 1988; Fram et al., 2011; McIntyre et al., 2003), although research suggests that many children are still aware of FI in the home, even if caregivers do not talk to them about it (Fram et al., 2011; Leung et al., 2020). More research is needed to understand why child versus caregiver reporting differences exist, including the role of enrollment in food assistance programs, which appeared to exacerbate reporting differences in one previous study (Landry et al., 2019a).

Regardless of the reason for caregiver and child reporting differences, child self-report of FI may more accurately reflect the child's own experience with FI and, thus, be more relevant to child behavioral and health outcomes. In this way, our findings align with another study where child-reported FI, but not mother-reported FI, was associated with child diet quality (Bernal et al., 2016). These findings suggest that future research on FI and children's behavioral and health

outcomes should measure child-reported FI rather than relying on caregiver report of household FI.

The salience of child perception of their ability to obtain enough quality food also suggests that traditional conceptualizations of EDrelated constructs such as dietary restraint may need to be reexamined. Dietary restraint, with or without caloric deprivation, is traditionally conceptualized as a psychological effort to resist certain food types or amounts of food in order to control weight, and is associated with more ED-related outcomes (Schaumberg & Anderson, 2016). It is conceivable that children with perceived FI may engage in behaviors that mirror dietary restraint, but not with the motivation of weight loss, that are under-captured in current assessment tools. Indeed, one study in adults with FI found that intentional dietary restraint for any reason (including reasons not related to weight control) was associated with more ED symptoms (Middlemass et al., 2021). Based on findings in the present study, we hypothesize that dietary restraint for any reason, including due to FI, may lead to food preoccupation, thus increasing the risk for other ED symptoms. However, it remains to be seen whether the consequences of restraint differ according to what is driving the behavior. This is an emerging area of research that requires more study.

We observed no gender differences in the ChEAT outcomes in bivariate analyses. This is consistent with other preadolescent samples (Bernier et al., 2010; Thomas et al., 2021) and suggests that gender differences in ED symptoms may be less prominent or less detectable in preadolescents compared to older samples. Although no measures of FI were associated with dieting (i.e., restricting food for weight control) in the full sample of children, there was evidence that these associations were modified by gender: caregiver-reported household FI was marginally associated with less dieting among girls only, and child-reported FI was associated with more dieting among boys only. To the best of our knowledge, no previous research has examined this association among preadolescent children only. One study of children aged 8-15 years found that household FI was associated with more weight loss attempts (Masler et al., 2021), but this study did not restrict to preadolescent children or stratify by gender, and the method of attempted weight loss was not specified. The association between FI and more dieting in boys only might be partially explained by parenting practices in the context of FI. Loth et al. (2014) found that problematic parental feeding practices like restrictive feeding and pressure to eat-which are more common among parents in food-insecure households (Bauer et al., 2015)-were associated with more dieting in boys but not girls. Future research should continue to investigate gender differences in the association between FI and dieting among preadolescents, including potential mechanisms for these gender differences.

This study has several strengths. We used detailed, well-validated measures for exposure and outcome variables, including measures of both caregiver- and child-reported FI. In addition, we focused on a sample of preadolescent children aged 8–10 years, which is a critical age period for ED prevention that has not been well studied regarding the association between FI- and ED-related outcomes. We also examined the novel and important issue of potential effect modification by gender.

This study does have some limitations. First, the cross-sectional study design precluded the establishment of temporality between

exposures and outcomes. It is conceivable that a child's experience of ED risk factors/symptoms could increase child and caregiver attention to FI in the home, increasing report of FI through reverse causation. This study also focused on low-income, English-speaking households from southeastern Michigan, United States, which may limit generalizability. The ChEAT-24 produces a total score that is difficult to interpret, as it combines risk factors/symptoms for different EDs with different presentations. We, therefore, incorporated the ChEAT-24 subscale scores as additional outcomes to better distinguish specific ED risk factors or symptoms. These subscales significantly enhanced interpretability of the findings, although we did not have a subscale for binge-eating behavior, which has been associated with FI in older populations (Hazzard et al., 2020). In addition, combining all caregivers of color into a single category may have obscured differences between racial/ethnic identities, since this category encompassed many different racial/ethnic groups: Black/ African American, Hispanic, Middle Eastern/North African, Asian, multi-racial/multi-ethnic, and other race/ethnicity. Similarly, use of binary gender options (boy or girl) may have resulted in incorrect gender labels for some children.

In conclusion, we found that child-reported FI, but not caregiver-reported household FI, was associated with ED-related outcomes among preadolescent boys and girls from low-income households. If our findings are confirmed by future research, then it is possible that FI and ED prevention/intervention programs would benefit from assessing and intervening on child self-report of FI. In addition to ensuring access to enough food for all children who report FI, it might also be helpful for programs to provide counseling to address the psychological impacts of FI in children. Finally, ED prevention/intervention programs could be targeted toward boys and girls in areas with a high prevalence of FI. Future studies should continue to investigate potential early life risk factors and mechanisms for the onset of EDs, especially in understudied groups. Specifically, large, prospective studies are needed to examine whether FI precedes the onset of ED risk factors/symptoms in preadolescents. Future research should collect data on both caregiver-reported household FI and child-reported FI and should assess well-defined ED risk factors/symptoms. Finally, future studies should continue to examine the role of potential moderators and mediators (e.g., gender, weight status, depressive symptoms, parental feeding practices, qualification for and enrollment in food assistance programs) for the association between FI and ED-related outcomes.

AUTHOR CONTRIBUTIONS

Mikayla R. Barry: Conceptualization; formal analysis; investigation; writing – original draft. Kendrin R. Sonneville: Conceptualization; methodology; supervision; writing – review and editing. Andrea R. McGowan: Writing – review and editing. Belinda L. Needham: Supervision; writing – review and editing. Lindsay C. Kobayashi: Supervision; writing – review and editing. Cindy W. Leung: Conceptualization; data curation; funding acquisition; methodology; supervision; writing – review and editing.

ACKNOWLEDGMENT

This study was supported by the Eunice Kennedy Shriver National Institute of Child Health and Human Development (grant no. 4R00HD084758).

CONFLICT OF INTEREST

The authors declare no conflict of interest.

DATA AVAILABILITY STATEMENT

The data that support the findings of this study are available from the corresponding author upon reasonable request.

ORCID

Mikayla R. Barry b https://orcid.org/0000-0003-3707-4823 Kendrin R. Sonneville b https://orcid.org/0000-0002-0359-3919 Belinda L. Needham b https://orcid.org/0000-0001-5939-2027 Lindsay C. Kobayashi b https://orcid.org/0000-0003-2725-3107 Cindy W. Leung b https://orcid.org/0000-0001-6718-3487

REFERENCES

- Altman, E. A., Ritchie, L. D., Frongillo, E. A., & Madsen, K. A. (2019). Food insecurity is associated with body dissatisfaction among children in California. *Journal of the Academy of Nutrition and Dietetics*, 119(10), 1732–1737. https://doi.org/10.1016/j.jand.2018.07.007
- Anderson, S. A. (1990). Core indicators of nutritional state for difficult-tosample populations. *The Journal of Nutrition*, 120(Suppl 11), 1559– 1600. https://doi.org/10.1093/jn/120.suppl_11.1555
- Barry, M. R., Sonneville, K. R., & Leung, C. W. (2021). Students with food insecurity are more likely to screen positive for an eating disorder at a large, public university in the Midwest. *Journal of the Academy of Nutrition and Dietetics*, 121(6), 1115–1124. https://doi.org/10.1016/j.jand.2021.01.025
- Bauer, K. W., MacLehose, R., Loth, K. A., Fisher, J. O., Larson, N. I., & Neumark-Sztainer, D. (2015). Eating- and weight-related parenting of adolescents in the context of food insecurity. *Journal of the Academy* of Nutrition and Dietetics, 115(9), 1408–1416. https://doi.org/10. 1016/j.jand.2015.01.011
- Becker, C. B., Middlemass, K. M., Gomez, F., & Martinez-Abrego, A. (2019). Eating disorder pathology among individuals living with food insecurity: A replication study. *Clinical Psychological Science*, 7(5), 1144– 1158. https://doi.org/10.1177/2167702619851811
- Bernal, J., Frongillo, E. A., & Rivera, J. A. (2016). Food insecurity reported by children, but not by mothers, is associated with lower quality of diet and shifts in foods consumed. *Maternal & Child Nutrition*, 12(3), 546–557. https://doi.org/10.1111/mcn.12206
- Bernard, R., Hammarlund, R., Bouquet, M., Ojewole, T., Kirby, D., Grizzaffi, J., & McMahon, P. (2018). Parent and child reports of food insecurity and mental health: Divergent perspectives. *The Ochsner Journal*, 18(4), 318–325. https://doi.org/10.31486/toj.18.0037
- Bernier, C. D., Kozyrskyj, A. L., Benoit, C., Becker, A. B., & Marchessault, G. (2010). Body image and dieting attitudes among preadolescents. *Canadian Journal of Dietetic Practice and Research*, 71(3), e34–e40. https:// doi.org/10.3148/71.3.2010.122
- Burke, M. P., Frongillo, E. A., Jones, S. J., Bell, B. B., & Hartline-Grafton, H. (2016). Household food insecurity is associated with greater growth in body mass index among female children from kindergarten through eighth grade. *Journal of Hunger and Environmental Nutrition*, 11(2), 227–241. https://doi.org/10.1080/19320248.2015.1112756
- Carlos Chavez, F. L., Hernandez, D. C., Harris, G. J., & Grzywacz, J. G. (2017). Household food security discordance among Latino adolescents and parents. *American Journal of Health Behavior*, 41(6), 775– 783. https://doi.org/10.5993/AJHB.41.6.11

- Coleman-Jensen, A., Rabbitt, M. P., Gregory, C. A., & Singh, A. (2021). Household food security in the United States in 2020, ERR-298. Washington, D.C.: Economic Research Service, U.S. Department of Agriculture.
- Combs, J. L., Pearson, C. M., Zapolski, T. C., & Smith, G. T. (2013). Preadolescent disordered eating predicts subsequent eating dysfunction. *Journal of Pediatric Psychology*, 38(1), 41–49. https://doi.org/10.1093/ jpepsy/jss094
- Conlon, B. A., McGinn, A. P., Lounsbury, D. W., Diamantis, P. M., Groisman-Perelstein, A. E., Wylie-Rosett, J., & Isasi, C. R. (2015). The role of parenting practices in the home environment among underserved youth. *Childhood Obesity*, 11(4), 394–405. https://doi.org/10. 1089/chi.2014.0093
- Department of Health and Human Services. (2019). Annual update of the HHS poverty guidelines (84 Fed. Reg. 1167). College Park, MD: Office of the Federal Register, U.S. National Archives and Records Administration.
- Economic Research Service. (2012). U.S. household food security survey module: Three-stage design, with screeners. Washington, D.C.: U.S. Department of Agriculture.
- El Zein, A., Shelnutt, K. P., Colby, S., Vilaro, M. J., Zhou, W., Greene, G., Olfert, M. D., Riggsbee, K., Morrell, J. S., & Mathews, A. E. (2019). Prevalence and correlates of food insecurity among U.S. college students: A multi-institutional study. *BMC Public Health*, 19(1), 660. https://doi.org/10.1186/s12889-019-6943-6
- Flagg, L. A., Sen, B., Kilgore, M., & Locher, J. L. (2014). The influence of gender, age, education and household size on meal preparation and food shopping responsibilities. *Public Health Nutrition*, 17(9), 2061– 2070. https://doi.org/10.1017/S1368980013002267
- Fitchen, J. M. (1988). Hunger, malnutrition, and poverty in the contemporary United States: Some observations on their social and cultural context. *Food Foodways*, 2(1), 309–333. https://doi.org/10.1080/ 07409710.1987.9961923
- Fram, M. S., Frongillo, E. A., Draper, C. L., & Fishbein, E. M. (2013). Development and validation of a child report assessment of child food insecurity and comparison to parent report assessment. *Journal of Hunger & Environmental Nutrition*, 8(2), 128–145. https://doi.org/10. 1080/19320248.2013.790775
- Fram, M. S., Frongillo, E. A., Jones, S. J., Williams, R. C., Burke, M. P., DeLoach, K. P., & Blake, C. E. (2011). Children are aware of food insecurity and take responsibility for managing food resources. *The Journal of Nutrition*, 141(6), 1114–1119. https://doi.org/10.3945/jn.110.135988
- Fram, M. S., Ritchie, L. D., Rosen, N., & Frongillo, E. A. (2015). Child experience of food insecurity is associated with child diet and physical activity. *The Journal of Nutrition*, 145(3), 499–504. https://doi.org/10. 3945/jn.114.194365
- Hanson, K. L., Sobal, J., & Frongillo, E. A. (2007). Gender and marital status clarify associations between food insecurity and body weight. *The Journal of Nutrition*, 137(6), 1460–1465. https://doi.org/10.1093/jn/ 137.6.1460
- Hazzard, V. M., Loth, K. A., Hooper, L., & Becker, C. B. (2020). Food insecurity and eating disorders: A review of emerging evidence. *Current Psychiatry Reports*, 22(12), 74. https://doi.org/10.1007/s11920-020-01200-0
- Hooper, L., Telke, S., Larson, N., Mason, S. M., & Neumark-Sztainer, D. (2020). Household food insecurity: Associations with disordered eating behaviours and overweight in a population-based sample of adolescents. *Public Health Nutrition*, 23(17), 3126–3135. https://doi.org/ 10.1017/S1368980020000464
- Hudson, J. I., Hiripi, E., Pope, H. G., Jr., & Kessler, R. C. (2007). The prevalence and correlates of eating disorders in the National Comorbidity Survey Replication. *Biological Psychiatry*, *61*(3), 348–358. https://doi. org/10.1016/j.biopsych.2006.03.040
- Jansen, E. C., Kasper, N., Lumeng, J. C., Brophy Herb, H. E., Horodynski, M. A., Miller, A. L., Contreras, D., & Peterson, K. E. (2017). Changes in household food insecurity are related to changes in BMI

and diet quality among Michigan head start preschoolers in a sex-specific manner. *Social Science* & *Medicine*, 1982(181), 168–176. https://doi.org/10.1016/j.socscimed.2017.04.003

- Jendrzyca, A., & Warschburger, P. (2016). Weight stigma and eating behaviours in elementary school children: A prospective population-based study. *Appetite*, 102, 51–59. https://doi.org/10.1016/j.appet.2016.02.005
- Kärkkäinen, U., Mustelin, L., Raevuori, A., Kaprio, J., & Keski-Rahkonen, A. (2018). Do disordered eating behaviours have long-term health-related consequences? *European Eating Disorders Review*, 26(1), 22–28. https://doi.org/10.1002/erv.2568
- Landry, M. J., van den Berg, A. E., Asigbee, F. M., Vandyousefi, S., Ghaddar, R., & Davis, J. N. (2019a). Child compared with parent perceptions of child-level food security. *Current Developments in Nutrition*, 3(10), nzz106. https://doi.org/10.1093/cdn/nzz106
- Landry, M. J., van den Berg, A. E., Asigbee, F. M., Vandyousefi, S., Ghaddar, R., & Davis, J. N. (2019b). Child-report of food insecurity is associated with diet quality in children. *Nutrients*, 11(7), 1574. https:// doi.org/10.3390/nu11071574
- Laraia, B., Vinikoor-Imler, L. C., & Siega-Riz, A. M. (2015). Food insecurity during pregnancy leads to stress, disordered eating, and greater postpartum weight among overweight women. *Obesity*, 23(6), 1303–1311. https://doi.org/10.1002/oby.21075
- Larsen, P. S., Strandberg-Larsen, K., Micali, N., & Andersen, A. M. (2015). Parental and child characteristics related to early-onset disordered eating: A systematic review. *Harvard Review of Psychiatry*, 23(6), 395– 412. https://doi.org/10.1097/HRP.000000000000073
- Lee, A. M., Scharf, R. J., & DeBoer, M. D. (2018). Association between kindergarten and first-grade food insecurity and weight status in U.S. children. *Nutrition*, 51-52, 1–5. https://doi.org/10.1016/j.nut.2017.12.008
- Leung, C. W., Stewart, A. L., Portela-Parra, E. T., Adler, N. E., Laraia, B. A., & Epel, E. S. (2020). Understanding the psychological distress of food insecurity: A qualitative study of children's experiences and related coping strategies. *Journal of the Academy of Nutrition and Dietetics*, 120(3), 395–403. https://doi.org/10.1016/j.jand.2019.10.012
- Lommi, S., Viljakainen, H. T., Weiderpass, E., & de Oliveira Figueiredo, R. A. (2020). Children's Eating Attitudes Test (ChEAT): A validation study in Finnish children. *Eating and Weight Disorders*, 25(4), 961–971. https:// doi.org/10.1007/s40519-019-00712-w
- Loth, K. A., MacLehose, R. F., Fulkerson, J. A., Crow, S., & Neumark-Sztainer, D. (2014). Are food restriction and pressure-to-eat parenting practices associated with adolescent disordered eating behaviors? *The International Journal of Eating Disorders*, 47(3), 310–314. https://doi. org/10.1002/eat.22189
- Maloney, M. J., McGuire, J. B., & Daniels, S. R. (1988). Reliability testing of a children's version of the Eating Attitude Test. *Journal of the American Academy of Child and Adolescent Psychiatry*, 27(5), 541–543. https:// doi.org/10.1097/00004583-198809000-00004
- Marques, E. S., Reichenheim, M. E., de Moraes, C. L., Antunes, M. M., & Salles-Costa, R. (2015). Household food insecurity: A systematic review of the measuring instruments used in epidemiological studies. *Public Health Nutrition*, 18(5), 877–892. https://doi.org/10.1017/ S1368980014001050
- Masler, I. V., Palakshappa, D., Skinner, A. C., Skelton, J. A., & Brown, C. L. (2021). Food insecurity is associated with increased weight loss attempts in children and adolescents. *Pediatric Obesity*, 16(1), e12691. https://doi.org/10.1111/ijpo.12691
- McClelland, J., Robinson, L., Potterton, R., Mountford, V., & Schmidt, U. (2020). Symptom trajectories into eating disorders: A systematic review of longitudinal, nonclinical studies in children/adolescents. *European Psychiatry*, 63(1), e60. https://doi.org/10.1192/j.eurpsy.2020.55
- McIntyre, L., Glanville, N. T., Raine, K. D., Dayle, J. B., Anderson, B., & Battaglia, N. (2003). Do low-income lone mothers compromise their nutrition to feed their children? CMAJ: Canadian Medical Association Journal, 168(6), 686–691.
- Middlemass, K. M., Cruz, J., Gamboa, A., Johnson, C., Taylor, B., Gomez, F., & Becker, C. B. (2021). Food insecurity & dietary restraint

EATING DISORDERS-WILEY

in a diverse urban population. *Eating Disorders*, *29*(6), 616–629. https://doi.org/10.1080/10640266.2020.1723343

- Middleton, G., Mehta, K., McNaughton, D., & Booth, S. (2018). The experiences and perceptions of food banks amongst users in high-income countries: An international scoping review. *Appetite*, 120, 698–708. https://doi.org/10.1016/j.appet.2017.10.029
- Murphy, T. J., Hwang, H., Kramer, M. S., Martin, R. M., Oken, E., & Yang, S. (2019). Assessment of eating attitudes and dieting behaviors in healthy children: Confirmatory factor analysis of the Children's Eating Attitudes Test. *The International Journal of Eating Disorders*, 52(6), 669–680. https://doi.org/10.1002/eat.23062
- Nagao-Sato, S., Druziako, S., Baltaci, A., Peralta Reyes, A. O., Zhang, Y., Hurtado Choque, G. A., & Reicks, M. (2021). Differences in reporting food insecurity and factors associated with differences among Latino fathers and mothers. *BMC Public Health*, 21(1), 912. https://doi.org/ 10.1186/s12889-021-10971-x
- Odoms-Young, A., & Bruce, M. A. (2018). Examining the impact of structural racism on food insecurity: Implications for addressing racial/ethnic disparities. *Family & Community Health*, 41, S3–S6. https://doi.org/10. 1097/FCH.00000000000183
- Poll, K. L., Holben, D. H., Valliant, M., & Joung, H. D. (2020). Food insecurity is associated with disordered eating behaviors in NCAA division 1 male collegiate athletes. *Journal of American College Health*, 68(2), 105–109. https://doi.org/10.1080/07448481.2018.1529035
- Rojo-Moreno, L., García-Miralles, I., Plumed, J., Barberá, M., Morales, M. M., Ruiz, E., & Livianos, L. (2011). Children's eating attitudes test: Validation in a sample of Spanish schoolchildren. *The International Journal of Eating Disorders*, 44(6), 540–546. https://doi.org/10.1002/eat.20855
- Schaumberg, K., & Anderson, D. (2016). Dietary restraint and weight loss as risk factors for eating pathology. *Eating Behaviors*, 23, 97–103. https://doi.org/10.1016/j.eatbeh.2016.08.009
- Smolak, L., & Levine, M. P. (1994). Psychometric properties of the Children's Eating Attitudes Test. *The International Journal of Eating Disorders*, 16(3), 275–282. https://doi.org/10.1002/1098-108x(199411)16: 3<275::aid-eat2260160308>3.0.co;2-u
- Stice, E., Marti, C. N., & Durant, S. (2011). Risk factors for onset of eating disorders: Evidence of multiple risk pathways from an 8-year prospective study. *Behaviour Research and Therapy*, 49(10), 622–627. https:// doi.org/10.1016/j.brat.2011.06.009
- Swanson, S. A., Horton, N. J., Crosby, R. D., Micali, N., Sonneville, K. R., Eddy, K., & Field, A. E. (2014). A latent class analysis to empirically describe eating disorders through developmental stages. *The International Journal of Eating Disorders*, 47(7), 762–772. https://doi.org/10.1002/eat.22308
- Tanofsky-Kraff, M., Shomaker, L. B., Stern, E. A., Miller, R., Sebring, N., Dellavalle, D., Yanovski, S. Z., Hubbard, V. S., & Yanovski, J. A. (2012). Children's binge eating and development of metabolic syndrome. *International Journal of Obesity*, 36(7), 956–962. https://doi.org/10.1038/ijo.2011.259
- Thomas, M., Miller, D. P., & Morrissey, T. W. (2019). Food insecurity and child health. *Pediatrics*, 144(4), e20190397. https://doi.org/10.1542/ peds.2019-0397
- Thomas, K. S., Williams, M. O., & Vanderwert, R. E. (2021). Disordered eating and internalizing symptoms in preadolescence. *Brain and Behavior*, 11(1), e01904. https://doi.org/10.1002/brb3.1904

How to cite this article: Barry, M. R., Sonneville, K. R., McGowan, A. R., Needham, B. L., Kobayashi, L. C., & Leung, C. W. (2022). Caregiver-reported household food insecurity and child-reported food insecurity in relation to eating disorder risk factors and symptoms among preadolescent children. *International Journal of Eating Disorders*, 55(10), 1331–1341. https://doi.org/10.1002/eat.23784